%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%
%%% Manuscript title - "Manifold-like matchbox manifolds"
%%%
%%% Author - Alex Clark, Steve Hurder & Olga Lukina
%%%
%%% Date Started: February 5, 2014
%%% Latest version: April 22, 2017
%%%
%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%


\documentclass{amsart}
\usepackage{graphicx}
\usepackage{amssymb}
\usepackage{epstopdf}
\usepackage{nicefrac}
\usepackage{enumerate}
\usepackage{hyperref}
\usepackage{float}
% \usepackage{color}
% \usepackage{pst-all}
\usepackage{tabularx}
\usepackage[usenames,dvipsnames,svgnames,table]{xcolor}
 
\input xy
\xyoption {all}

\DeclareGraphicsRule{.tif}{png}{.png}{`convert #1 `dirname #1`/`basename #1 .tif`.png}

\parskip = 6pt
\parindent = 0.0in

\hoffset=-.7in
\voffset=-.7in
\setlength{\textwidth}{6in}
\setlength{\textheight}{9.1in}
\setlength{\marginparwidth}{0.7in}

\newtheorem{thm}{THEOREM}[section]
\newtheorem{conj}[thm]{CONJECTURE}
\newtheorem{cor}[thm]{COROLLARY}
\newtheorem{defn}[thm]{DEFINITION}
\newtheorem{ex}[thm]{EXAMPLE}
\newtheorem{hyp}[thm]{HYPOTHESES}
\newtheorem{lemma}[thm]{LEMMA}
\newtheorem{prob}[thm]{PROBLEM}
\newtheorem{prop}[thm]{PROPOSITION}
\newtheorem{quest}[thm]{QUESTION}
\newtheorem{remark}[thm]{REMARK}
\newtheorem{conjecture}[thm]{CONJECTURE}

% math notation shortcuts
\newcommand\mor{\mathrel{\stackrel{\makebox[0pt]{\mbox{\normalfont\tiny Ret}}}{\sim}}}
\newcommand\myeq{\mathrel{\stackrel{\makebox[0pt]{\mbox{\normalfont\tiny def}}}{=}}}
\newcommand\homeo{\mathrel{\stackrel{\makebox[0pt]{\mbox{\normalfont\tiny top}}}{\approx}}}
\newcommand{\ds}{\displaystyle}

 

\newcommand{\F}{{\mathcal F}} % our favorite foliation
\newcommand{\cA}{{\mathcal A}}
\newcommand{\cB}{{\mathcal B}}
\newcommand{\cC}{{\mathcal C}}
\newcommand{\cG}{{\mathcal G}}
\newcommand{\cGF}{\cG_{\F}} % denotes the pseudogroup of F
\newcommand{\cH}{{\mathcal H}}
\newcommand{\cI}{{\mathcal I}}
\newcommand{\cJ}{{\mathcal J}}
\newcommand{\cK}{{\mathcal K}}
\newcommand{\cL}{{\mathcal L}}
\newcommand{\cM}{{\mathcal M}}
\newcommand{\cO}{{\mathcal O}}
\newcommand{\cP}{{\mathcal P}}
\newcommand{\cR}{{\mathcal R}}
\newcommand{\cS}{{\mathcal S}}
\newcommand{\cT}{{\mathcal T}}
\newcommand{\cU}{{\mathcal U}}
\newcommand{\cV}{{\mathcal V}}
\newcommand{\cW}{{\mathcal W}}
\newcommand{\cY}{{\mathcal Y}}
\newcommand{\dF}{d_{\F}} % leafwise metric
\newcommand{\dFU}{\delta^{\F}_{\cU}} 
\newcommand{\diam}{{\rm diam}} %
\newcommand{\dM}{d_{\fM}} %
\newcommand{\dT}{d_{\cT}} %
\newcommand{\dTi}{d_{\cT_i}} %
\newcommand{\dX}{d_{\fX}} %
\newcommand{\Dom}{{\rm Dom}} %
\newcommand{\e}{{\epsilon}} % some choice
\newcommand{\eM}{{\epsilon_{\fM}}}
\newcommand{\fC}{{\mathfrak{C}}}
\newcommand{\fD}{{\mathfrak{D}}}
\newcommand{\fF}{\mathfrak{F}}
\newcommand{\fG}{\mathfrak{G}}
\newcommand{\fM}{{\mathfrak{M}}}
\newcommand{\fR}{{\mathfrak{R}}}
\newcommand{\fT}{{\mathfrak{T}}}
\newcommand{\fU}{{\mathfrak{U}}}
\newcommand{\fX}{{\mathfrak{X}}}
\newcommand{\Homeo}{{\rm Homeo}}  
\newcommand{\lF}{{\lambda_{\mathcal F}}}
\newcommand{\mD}{{\mathbb D}}
\newcommand{\mN}{{\mathbb N}}
\newcommand{\mR}{{\mathbb R}}
\newcommand{\mS}{{\mathbb S}}
\newcommand{\mT}{{\mathbb T}}
\newcommand{\mZ}{{\mathbb Z}}
\newcommand{\oU}{{\overline{U}}}
\newcommand{\vp}{{\varphi}}
\newcommand{\whPhi}{{\widehat{\Phi}}}
\newcommand{\whU}{{\widehat U}}
\newcommand{\whvarp}{{\widehat \varphi}}
\newcommand{\wtB}{\widetilde{B}}
\newcommand{\wtcH}{{\widetilde{\mathcal H}}}
\newcommand{\wtgamma}{{\widetilde{\gamma}}}
\newcommand{\wtPhi}{{\widetilde{\Phi}}}
\newcommand{\wtpi}{{\widetilde{\pi}}}
\newcommand{\wtY}{\widetilde{Y}}

\newcommand{\rp}{\rho_{\pi}} % modulus of continuity function for projections defined in equation~\ref{eq-modpi}
\newcommand{\rt}{\rho_{\tau}} % modulus of continuity function for sections defined in equation~\ref{eq-modtau}

 
\newcommand{\psg}{{\rm pseudo}{\star}{\rm group}}

\newcommand{\ef}{{\epsilon_{f}}}    
\newcommand{\eF}{{\epsilon_{\F}}}    
\newcommand{\eU}{{\epsilon_{\cU}}} % Lebesgue number for given covering.
 

\begin{document}
\begin{thebibliography}{10}

\bibitem{AM1988}
{J.M.~Aarts and M.~Martens},
\newblock {\it Flows on one-dimensional spaces},
\newblock {\bf Fund. Math.}, 131:39--58, 1988.

\bibitem{AO1991}
{J.~Aarts and L.~Oversteegen},
\newblock {\it Flowbox manifolds},
\newblock {\bf Trans. Amer. Math. Soc.}, 327:449--463, 1991.

\bibitem{AHO1991}
{J.M.~Aarts, C.L.~Hagopian and L.G.~Oversteegen},
\newblock {\it The orientability of matchbox manifolds},
\newblock {\bf Pacific. J. Math.}, 150:1--12, 1991.


\bibitem{AO1995}
{J.~Aarts and L.~Oversteegen},
\newblock {\it Matchbox manifolds},
\newblock In {\bf Continua ({C}incinnati, {OH}, 1994)},
\newblock {Lecture Notes in Pure and Appl. Math., Vol. 170},
\newblock {Dekker, New York}, 1995, pages 3--14..

 

\bibitem{Alexandroff1928}
{P.~Alexandroff},
\newblock {\it Untersuchungen \"{u}ber Gestalt und Lage abgeschlossener Mengen beliebiger Dimension,}
\newblock {\bf Ann. of Math.}, 30:101--187, 1928/29.

 
 

\bibitem{Bing1960}
{R.H.~Bing},
\newblock {\it A simple closed curve is the only homogeneous bounded
plane continuum that contains an arc},
\newblock {\bf Canad. J. Math.}, 12:209--230, 1960.

\bibitem{BC1964}
{R.~Bishop and R.~Crittenden},
\newblock {\bf Geometry of manifolds},
\newblock {(reprint of the 1964 original),}
\newblock AMS Chelsea Publishing, Providence, RI, 2001.




\bibitem{CandelConlon2000}
{A.~Candel and L.~Conlon},
\newblock {\bf Foliations I},
\newblock Amer. Math. Soc., Providence, RI, 2000.


\bibitem{ClarkHurder2013}
{A.~Clark and S.~Hurder},
\newblock {\it Homogeneous matchbox manifolds},
\newblock {\bf Trans. A.M.S}, 365:3151--3191, 2013.

 
\bibitem{CHL2014}
{A.~Clark, S.~Hurder and O.~Lukina},
\newblock {\it Shape of matchbox manifolds},
\newblock {\bf Indag. Math.}, 25(4):669--712, 2014.

\bibitem{CHL2013c}
{A.~Clark, S.~Hurder and O.~Lukina},
\newblock {\it Classifying matchbox manifolds},
\newblock {preprint}, August 2013, {arXiv:1311.0226}.


         
\bibitem{Deleanu1962}
{A.~Deleanu}, 
\newblock {\it On spaces which may be mapped with arbitrarily small counter-images onto manifolds}, 
\newblock {\bf Bull. Acad. Polon. Sci. S\'er. Sci. Math. Astronom. Phys.}, 10:193--198, 1962.

        
\bibitem{Deleanu1963}
{A.~Deleanu}, 
\newblock {\it On {$\varepsilon$}-maps of polyhedra onto manifolds}, 
\newblock {\bf Michigan Math. J.}, 10:363--364, 1963.

  
 
\bibitem{doCarmo1992}
{M.~do~Carmo},
\newblock {\bf Riemannian geometry},
\newblock {Translated from the second Portuguese edition by Francis Flaherty},
\newblock {Birkh\"auser Boston Inc.}, {Boston}, 1992.

\bibitem{DHL2016a}
{J.~Dyer, S.~Hurder and O.~Lukina},
\newblock {\it The discriminant invariant of Cantor group actions},
\newblock {\bf Topology Appl.}, 208: 64-92, 2016;  {arXiv:1509.06227}


\bibitem{DHL2016b}
{J.~Dyer, S.~Hurder and O.~Lukina},
\newblock {\it Growth and homogeneity of matchbox manifolds},
\newblock {\bf Indag. Math.}, 28:145--169, 2017;  {arXiv:1602.00784}.

\bibitem{DHL2016c}
{J.~Dyer, S.~Hurder and O.~Lukina},
\newblock {\it Molino theory for matchbox manifolds},
\newblock {\bf Pacific J. Math.}, to appear;  {arXiv:1610.03896}.

              
\bibitem{Edwards1980}
{R.D.~Edwards},
\newblock {\it The topology of manifolds and cell-like maps},
\newblock in {\bf Proceedings of the {I}nternational {C}ongress of {M}athematicians ({H}elsinki, 1978)}, 
\newblock {Acad. Sci. Fennica, Helsinki}, 1980, pages 111--127.
     
              
\bibitem{Eilenberg1938}
{S.~Eilenberg},
\newblock {\it Sur les transformations \`a petite tranches},
\newblock {\bf Fund. Math},  30:92--95, 1938. 
     
     

 
\bibitem{Ganea1959}
{T.~Ganea},
\newblock {\it On {$\varepsilon$}-maps onto manifolds},
\newblock {\bf Fund. Math.}  47:35--44, 1959.
           
\bibitem{Ghys1999}
{\'{E}~Ghys},
\newblock {\it Laminations par surfaces de Riemann},
\newblock in {\bf Dynamique et G\'{e}om\'{e}trie Complexes},
\newblock {Panoramas \& Synth\`{e}ses}, 8:49--95, 1999.


\bibitem{Hagopian1977}
{C.~Hagopian},
\newblock {\it A characterization of solenoids},
\newblock {\bf Pacific J. Math.}, 68:425--435, 1977.

   
 
\bibitem{Hurder2014}
{S.~Hurder},
\newblock {\it Lectures on Foliation Dynamics},
\newblock {\bf Foliations: Dynamics, Geometry and Topology}, pages 87--149,
\newblock {Advanced Courses in Mathematics CRM Barcelona}, Springer Basel,   2014.

 \bibitem{HL2017a}
{S.~Hurder and O.~Lukina},
\newblock {\it Wild solenoids},
\newblock {\it submitted},    {arXiv:1702.03032}.

\bibitem{LM1997}
{M.~Lyubich and Y.~Minsky},
\newblock {\it Laminations in holomorphic dynamics},
\newblock {\bf J. Differential Geom.}, 47:17--94, 1997.


      
\bibitem{MardesicSegal1963}
{S.~Marde\v{s}i\'{c} and J.~Segal},
\newblock {\it  $\e$--mappings onto polyhedra},
\newblock {\bf Trans. Amer. Math. Soc.}, 109:146--164, 1963.

\bibitem{MardesicSegal1967}
{S.~Marde\v{s}i\'{c} and J.~Segal},
\newblock {\it  {$\epsilon $}-mappings and generalized manifolds},
\newblock {\bf Michigan Math. J.}, 14:171--182, 1967.


 \bibitem{Matsumoto2010}
 {S.~Matsumoto},
 \newblock {\it The unique ergodicity of equicontinuous laminations}, 
\newblock {\bf Hokkaido Math. J.}, 39(3): 389-403, 2010.


\bibitem{McCord1965}
{M.C.~McCord},
\newblock {\it Inverse limit sequences with covering maps},
\newblock {\bf Trans. Amer. Math. Soc.}, 114:197--209, 1965

     
\bibitem{MR1989}
{M.~Mislove and J.T.~Rogers, Jr.},
\newblock {\it Local product structures on homogeneous continua},
\newblock {\bf Topology Appl.},31:259--267, 1989.



\bibitem{MS2006}
{C.C.~Moore and C.~Schochet},
\newblock {\bf Analysis on Foliated Spaces},
\newblock Math. Sci. Res. Inst. Publ. vol. 9, Second Edition,
\newblock Cambridge University Press, New York, 2006.



\bibitem{Sullivan2014}
{D.~Sullivan},
\newblock{Solenoidal manifolds},
 \newblock {\bf J. Singul.}, 9:203--205, 2014.

 
\bibitem{EThomas1973}
{E.S.~Thomas, Jr.}
\newblock {\it One-dimensional minimal sets},
\newblock {\bf Topology}, 12:233--242, 1973.


   
\bibitem{vanDantzig1930}
{D.~van Dantzig},
\newblock {\it  \"{U}ber topologisch homogene {K}ontinua},
\newblock {\bf Fund. Math.}, 15:102--125, 1930.

\bibitem{Verjovsky2014}
{A.~Verjovsky},
\newblock{Commentaries on the paper {\it {S}olenoidal manifolds} by  {D}ennis {S}ullivan},
 \newblock {\bf J. Singul.}, 9:245--251, 2014.

     
\bibitem{Vietoris1927}
{L.~ Vietoris},
\newblock {\it  \"{U}ber den h\"oheren {Z}usammenhang kompakter {R}\"aume und eine {K}lasse von zusammenhangstreuen {A}bbildungen},
\newblock {\bf Math. Ann.}, 97:454--472, 1927.
 
  
\end{thebibliography}


% \pagecolor{black}
% \color{red}

\title{Manifold--like matchbox manifolds}

\thanks{2010 {\it Mathematics Subject Classification}. Primary 57N25,37B45; Secondary 54F15 }

\author{Alex Clark}
% \thanks{AC and OL supported in part by EPSRC grant EP/G006377/1}
\address{Alex Clark, Department of Mathematics, University of Leicester, University Road, Leicester LE1 7RH, United Kingdom}
\email{Alex.Clark@leicester.ac.uk}

\author{Steven Hurder}
\address{Steven Hurder, Department of Mathematics, University of Illinois at Chicago, 322 SEO (m/c 249), 851 S. Morgan Street, Chicago, IL 60607-7045}
\email{hurder@uic.edu}

\author{Olga Lukina}
\address{Olga Lukina, Department of Mathematics, University of Illinois at Chicago, 322 SEO (m/c 249), 851 S. Morgan Street, Chicago, IL 60607-7045}
\email{lukina@uic.edu}

 \thanks{Version date: April 22, 2017}

\date{}

% \subjclass{Primary 54F15, 54C56, 37B45, 57R30; Secondary , 57R65}

  \begin{abstract}
  A matchbox manifold is a generalized lamination, which is a continuum whose arc-components define the leaves of a foliation of the space.
The main result of this paper    implies that a matchbox manifold which is manifold-like must be homeomorphic to a weak solenoid. 
  \end{abstract}

\maketitle

 
 
 
 % \colorbox{BurntOrange}

\section{Introduction} \label{sec-intro}

A \emph{continuum} is a compact, connected, and non-empty
metrizable space. The notion of a manifold-like continuum is derived from the notion of an $\e$-map, which was introduced by 
   Alexandroff     \cite{Alexandroff1928}: 
\begin{defn}    \label{def-Ylike}
 Let  $X$ be a metric space, $Y$ a topological space and $\e > 0$ a constant. Then a   map $f \colon X \to Y$ is  said to be an $\e$-map if $f$ is a continuous surjection and for each point $y\in Y$,   the inverse image $f^{-1}(y)$   has diameter less than $\e$.   A metric space $X$ is said to be \emph{$Y$--like}, for some topological space $Y$, if for every   $\e>0$, there is an $\e$-map $f_\e \colon X \to Y$.
 \end{defn}
 
 
    
 
For example, a space $X$ is \emph{circle-like} if it is $Y$-like, where $Y = \mS^1$ is the circle. 
More generally, let 
\begin{equation}
\cM(n)=\{M\,|\, M \text{ is a closed connected manifold of dimension }n\,\}.
\end{equation}
\begin{defn}  \label{def-Mlike}
 A continuum  $X$ is said to be   \emph{manifold-like}, if there exists $n \geq 1$ such that for every   $\e>0$, there exists $M_{\e} \in \cM(n)$ and an $\e$-map   $f_\e \colon X \to M_{\e}$.
 \end{defn}

The study of the properties of $\e$-maps and   manifold-like continua has a long history in the study  of the topology of spaces.   Eilenberg showed in \cite{Eilenberg1938} that an $\e$-map, for $\e > 0$ sufficiently small,   admits a left approximate inverse.
 Ganea  studied the properties of  compact, locally connected  manifold-like ANR's of dimension $n$ in \cite{Ganea1959}, and showed that such a space  has the homotopy type of a closed $n$-manifold. 
Deleanu \cite{Deleanu1962,Deleanu1963} showed that an $n$-dimensional connected polyhedron which is manifold-like is a closed pseudo-manifold.
Bob Edwards gave in his 1978  ICM address \cite{Edwards1980}  an overview of the further applications of $\e$-approximations and homeomorphisms. 


Marde\v{s}i\'{c} and Segal    \cite{MardesicSegal1963,MardesicSegal1967} studied the properties of manifold-like connected polyhedron, and gave conditions under which such spaces must be a topological manifold. These   authors used a technique of approximation of the given continuum by an inverse limit of spaces, and noted that their results do not apply to a continuum which is not locally connected, such as the   dyadic solenoid. 

 The goal of this work is to characterize a class of manifold-like continua for which the Marde\v{s}i\'{c} and Segal results do not apply. These are the matchbox manifolds as studied by the authors in 
 \cite{ClarkHurder2013,CHL2014,CHL2013c}, and discussed below. Our study of matchbox manifolds in these works was inspired by a result of Bing in \cite{Bing1960}.
 Recall that  a topological  space $X$ is \emph{homogeneous} if for
every $x, y \in X$, there exists a homeomorphism $h \colon X \to X$ such that $h(x) = y$. 
\begin{thm}  \label{thm-bing}
Let $X$ be    a homogeneous, circle-like
continuum that contains an arc. Then either $X$ is homeomorphic to the circle $\mS^1$, or to    an inverse limit of coverings of   $\mS^1$.
\end{thm}
This results inspired the subsequent works  by McCord \cite{McCord1965}, Thomas \cite{EThomas1973}, Hagopian \cite{Hagopian1977}, Mislove and Rogers \cite{MR1989}, and Aarts, Hagopian and Oversteegen \cite{AHO1991}, all for $1$-dimensional flow spaces. 

In this work, we give extensions of Theorem~\ref{thm-bing} to continua with higher dimensional arc-components. We first recall two notions which are required to formulate our results.  A weak solenoid $\cS_\cP$ is the inverse limit space of a sequence of   covering maps of finite degree greater than one,
\begin{equation}\label{eq-presentation}
\cP = \{p_{\ell +1} \colon M_{\ell +1} \to M_{\ell} \mid \ell \geq 0\},
\end{equation}
   where $M_\ell$ is a compact connected manifold without boundary. The collection of maps $\cP$ is called a \emph{presentation} for $\cS_\cP$.  
  A weak solenoid $\cS_\cP$ is \emph{regular} if the presentation $\cP$ can be chosen so that for each $\ell \geq 0$, the composition $p_{\ell}^0 \equiv p_1 \circ \cdots p_{\ell} \colon M_\ell \to M_0$ is a regular covering map; that is, the fundamental group of $M_\ell$
injects onto a normal subgroup of the fundamental group of $M_0$ under the
map induced by the covering projection $p_{\ell}^0$.   A weak solenoid which is not regular is said to be \emph{irregular}.   A \emph{Vietoris solenoid} \cite{vanDantzig1930,Vietoris1927}  is   a $1$-dimensional  regular solenoid, where each $M_{\ell}$ is a circle, as arises in the conclusion of Theorem~\ref{thm-bing}.
   

A matchbox manifold    $\fM$ is a continuum  equipped with a decomposition $\F$ into leaves of constant dimension, 
 so that the pair $(\fM, \F)$ is a foliated space  in the sense of \cite{MS2006}, for which the local transversals to the foliation  are totally disconnected. In particular,    
the leaves of $\F$ are the path connected components of $\fM$.   A matchbox manifold with $2$-dimensional leaves is a lamination by surfaces in the sense of Ghys  \cite{Ghys1999} and Lyubich and Minsky \cite{LM1997}, while Sullivan called them ``solenoidal spaces'' in \cite{Sullivan2014,Verjovsky2014}. The terminology ``matchbox manifold'' follows  the usage introduced in  \cite{AM1988,AO1991,AO1995}. 
    A Vietoris solenoid is a $1$-dimensional matchbox manifold, and more generally, McCord showed in \cite{McCord1965} that 
  $n$-dimensional solenoids   are examples of $n$-dimensional  matchbox manifolds.
 
 Next, recall a result of the first two authors. A matchbox manifold is said to be \emph{equicontinuous} if the holonomy pseudogroup $\cGF$ associated to the foliation $\F$ (see Section~\ref{sec-holonomy}) defines an equicontinuous  action on its transversal space, as defined in Definition~\ref{def-equicontinuous}.
 \begin{thm} \cite[Theorems~1.2 and 1.4]{ClarkHurder2013} \label{thm-CH2013}
Let $\fM$ be an equicontinuous   matchbox manifold. Then $\fM$ is
homeomorphic to a  weak solenoid, and in particular is manifold-like.  Moreover, if $\fM$ is homogeneous, then $\fM$ is homeomorphic to a regular solenoid.
\end{thm}
 
 The main result  of  this paper, as follows, yields a generalization of   Theorem~\ref{thm-bing} to higher dimensional matchbox manifolds. 
\begin{thm}\label{thm-main1}
A manifold-like matchbox manifold  $\fM$ is equicontinuous.
\end{thm}

  Theorems~\ref{thm-CH2013} and \ref{thm-main1} then yield the following  partial    converse to Theorem~\ref{thm-CH2013}:

\begin{cor}\label{cor-weak}
A    manifold-like matchbox manifold $\fM$ is homeomorphic to a  weak  solenoid. 
\end{cor}

The hypothesis that a matchbox manifold $\fM$ is manifold-like does not imply that $\fM$ is homogeneous, as the ``discriminant obstruction'' to homogeneity for a weak solenoid is supported in arbitrarily small open neighborhoods of points in $\fM$ . The discriminant invariant was introduced and studied in the works in \cite{DHL2016a,DHL2016b,DHL2016c,HL2017a}.
  
The remainder of this paper is organized as follows. In Section~\ref{sec-concepts} we recall the definitions of foliated spaces and matchbox manifolds, and give some of their basic properties. Particular care is taken to introduce various metric estimates related to the geometry of the leaves of the foliation, and to its dynamical properties. 
In Section~\ref{sec-holonomy} we recall the construction of the holonomy along leafwise paths. 

In Section~\ref{sec-pathlifting}, we prove Theorem~\ref{thm-main1}, using the path lifting property for $\e$-maps from a matchbox manifold to a compact manifold. This is the key technical tool, which is used to show that the foliation $\F$ on $\fM$ must be equicontinuous.   

The Appendix~\ref{appendix-A} contains the proof of a technical result, Proposition~\ref{prop-coverage} below. The proof of this result does not appear to be in the literature, and may even have a simpler proof than is given here. However, the result is essential for the proof of  Theorem~\ref{thm-main1}, so is included for completeness.
 
\eject

\section{Foliated spaces and matchbox manifolds} \label{sec-concepts}

We recall some   background concepts used in the proof of our main theorems.  

\subsection{Matchbox manifolds}\label{subsec-fs} 
We first recall the definition  of a  matchbox manifold.

\begin{defn} \label{def-fs}
A \emph{matchbox manifold of dimension $n$} is a   continuum $\fM$, such that  there exists a compact, separable, totally disconnected metric space $\fX$, and
for each $x \in \fM$ there is a compact subset $\fT_x \subset \fX$, an open subset $U_x \subset \fM$, and a homeomorphism $\vp_x \colon \oU_x \to [-1,1]^n \times \fT_x$ defined on the closure $\oU_x$ in $\fM$,
  such that $\vp_x(x) = (0, w_x)$ where $w_x \in int(\fT_x)$. Moreover, it is assumed that each $\vp_x$  admits an extension to a foliated homeomorphism
$\whvarp_x \colon \whU_x \to (-2,2)^n \times \fT_x$ where $\whU_x \subset \fM$ is an open subset such that $\oU_x \subset \whU_x$.
The space $\fT_x$   is  called the \emph{local transverse model} at $x$.  
\end{defn}

The assumption that the transversals $\fT_x$ are totally disconnected implies that the local charts $\vp_x$ satisfy the compatibility axioms of foliation charts for a foliated space, as   in  \cite{CandelConlon2000,ClarkHurder2013,MS2006}. 

 

Let $\pi_x \colon \oU_x \to \fT_x$ denote the composition of $\vp_x$ with projection onto the second factor.

Also introduce the transversal maps $\tau_x \colon \fT_x \to \cT_x \subset \fM$, defined for $w \in \fT_x$ by $\tau_x(w) = \vp_x^{-1}(0 , w)$.
The subspace $\cT_x$ is   given the   metric $d_{\cT_x}$ which is the restriction of  the metric $\dM$.  

 
For $w \in \fT_x$ the set $\cP_x(w) = \pi_x^{-1}(w) \subset \oU_x$ is called a \emph{plaque} for the coordinate chart $\vp_x$. We adopt the notation, for $z \in \oU_x$, that $\cP_x(z) = \cP_x(\pi_x(z))$, so that $z \in \cP_x(z)$. Note that each plaque $\cP_x(w)$ is given the topology so that the restriction $\vp_x \colon \cP_x(w) \to [-1,1]^n \times \{w\}$ is a homeomorphism. Then $int (\cP_x(w)) = \vp_x^{-1}((-1,1)^n \times \{w\})$.

Let $U_x = int (\oU_x) = \vp_x^{-1}((-1,1)^n \times int(\fT_x))$.
Note that if $z \in U_x \cap U_y$, then $int(\cP_x(z)) \cap int( \cP_y(z))$ is an open subset of both
$\cP_x(z) $ and $\cP_y(z)$.
The collection of sets
$$\cV = \{ \vp_x^{-1}(V \times \{w\}) \mid x \in \fM ~, ~ w \in \fT_x ~, ~ V \subset (-1,1)^n ~ {\rm open}\}$$
forms the basis for the \emph{fine topology} of $\fM$. The connected components of the fine topology are called leaves, and define the foliation $\F$ of $\fM$.
 In particular,    the leaves of the foliation $\F$ of $\fM$  are the path-connected components of $\fM$.
For $x \in \fM$, let $L_x \subset \fM$ denote the leaf of $\F$ containing $x$.

\begin{defn} \label{def-sfs}
A \emph{smooth matchbox manifold} is a   space $\fM$ as above, such that there exists a choice of local charts $\vp_x \colon \oU_x \to [-1,1]^n \times \fT_x$ such that for all $x,y \in \fM$ with $z \in U_x \cap U_y$, there exists an open set $z \in V_z \subset U_x \cap U_y$ such that $\cP_x(z) \cap V_z$ and $\cP_y(z) \cap V_z$ are connected open sets, and the composition
$$\psi_{x,y;z} \equiv \vp_y \circ \vp_x ^{-1}\colon \vp_x(\cP_x (z) \cap V_z) \to \vp_y(\cP_y (z) \cap V_z)$$
is a smooth map, where $\vp_x(\cP_x (z) \cap V_z) \subset \mR^n \times \{w\} \cong \mR^n$ and $\vp_y(\cP_y (z) \cap V_z) \subset \mR^n \times \{w'\} \cong \mR^n$. The leafwise transition maps $\psi_{x,y;z}$ are assumed to depend continuously on $z$ in the $C^{\infty}$-topology.
\end{defn}

A map $f \colon \fM \to \mR$ is said to be \emph{smooth} if for each flow box
$\vp_x \colon \oU_x \to [-1,1]^n \times \fT_x$ and $w \in \fT_x$ the composition
$y \mapsto f \circ \vp_x^{-1}(y, w)$ is a smooth function of $y \in (-1,1)^n$, and depends continuously on $w$ in the $C^{\infty}$-topology on maps of the plaque coordinates $y$. As noted in \cite{MS2006} and \cite[Chapter 11]{CandelConlon2000}, this allows one to define smooth partitions of unity, vector bundles, and tensors for smooth foliated spaces. In particular, one can define leafwise Riemannian metrics. We recall a standard result, whose proof for foliated spaces can be found in \cite[Theorem~11.4.3]{CandelConlon2000}.
\begin{thm}\label{thm-riemannian}
Let $\fM$ be a smooth matchbox manifold. Then there exists a leafwise Riemannian metric for $\F$, such that for each $x \in \fM$, the leaf $L_x$ inherits the structure of a complete Riemannian manifold with bounded geometry, and the Riemannian metric and its covariant derivatives depend continuously on $x$ .  
\end{thm}
Bounded geometry implies, for example, that for each $x \in \fM$, there is a leafwise exponential map
$\exp^{\F}_x \colon T_x\F \to L_x$ which is a surjection, and the composition $\exp^{\F}_x \colon T_x\F \to L_x \subset \fM$ depends continuously on $x$ in the compact-open topology on maps.
 All matchbox manifolds  are assumed to be smooth with a given leafwise Riemannian metric, and with a fixed choice of    metric $\dM$ on $\fM$.
 

\subsection{Metric estimates}\label{subsec-metric}
We formulate some relations between the metric properties of a matchbox manifold $\fM$ and the metric properties of the leaves of $\F$. These technical conditions are used in studying the dynamics and geometry of these spaces.
 

For $x \in \fM$ and $\e > 0$, let $D_{\fM}(x, \e) = \{ y \in \fM \mid \dM(x, y) \leq \e\}$ be the closed $\e$-ball about $x$ in $\fM$, and $B_{\fM}(x, \e) = \{ y \in \fM \mid \dM(x, y) < \e\}$ the open $\e$-ball about $x$.

Recall that $\dX$ denotes the metric on the   space $\fX$ in Definition~\ref{def-mm}.  For $w \in \fX$ and $\e > 0$, let $D_{\fX}(w, \e) = \{ w' \in \fX \mid \dX(w, w') \leq \e\}$ be the closed $\e$-ball about $w$ in $\fX$, and let $B_{\fX}(w, \e) = \{ w' \in \fX \mid \dX(w, w') < \e\}$ be the open $\e$-ball about $w$.

Each leaf $L \subset \fM$ has a complete path-length metric, induced from the leafwise Riemannian metric:
$$\dF(x,y) = \inf \left\{\| \gamma\| \mid \gamma \colon [0,1] \to L ~{\rm is ~ piecewise ~~ C^1}~, ~ \gamma(0) = x ~, ~ \gamma(1) = y ~, ~ \gamma(t) \in L \quad \forall ~ 0 \leq t \leq 1\right\}$$
  where $\| \gamma \|$ denotes the path-length of the piecewise $C^1$-curve $\gamma(t)$. If $x,y \in \fM$   are not on the same leaf, then set $\dF(x,y) = \infty$.

  For each $x \in \fM$ and $r > 0$, let $D_{\F}(x, r) = \{y \in L_x \mid \dF(x,y) \leq r\}$.

 
 

For each $x \in \fM$, the  {Gauss Lemma} implies that there exists $\lambda_x > 0$ such that $D_{\F}(x, \lambda_x)$ is a \emph{strongly convex} subset for the metric $\dF$. That is, for any pair of points $y,y' \in D_{\F}(x, \lambda_x)$ there is a unique shortest geodesic segment in $L_x$ joining $y$ and $y'$ and  contained in $D_{\F}(x, \lambda_x)$. This standard concept of Riemannian geometry is discussed in detail in  \cite{BC1964}, and in \cite[Chapter 3, Proposition 4.2]{doCarmo1992}. 
Then for all $0 < \lambda < \lambda_x$ the disk $D_{\F}(x, \lambda)$ is also strongly convex. 
The leafwise metrics for $\F$  constructed in the proof of Theorem~\ref{thm-riemannian} have uniformly bounded geometry, and the  first and second order covariant derivatives of the metrics depend continuously on the point $x \in \fM$, so by the compactness of $\fM$, we obtain:
\begin{lemma}\label{lem-stronglyconvex}
There exists $\lF > 0$ such that for all $x \in \fM$, $D_{\F}(x, \lF)$ is strongly convex.
\end{lemma}

 
  
\subsection{Regular coverings} \label{subsec-rc}
We next formulate  the definition of a \emph{regular covering} of a matchbox manifold $\fM$.
It follows from standard considerations (see \cite{ClarkHurder2013}) that a matchbox manifold admits a covering by foliation charts which satisfies additional regularity conditions.

\begin{prop}\label{prop-regular}\cite{ClarkHurder2013}
For a smooth foliated space $\fM$, given $\eM > 0$, there exist   $\lF>0$ and a choice of local charts $\vp_x \colon \oU_x \to [-1,1]^n \times \fT_x$ with the following properties:  For each $x \in \fM$,
\begin{enumerate}
\item  $U_x \equiv int(\oU_x) = \vp_x^{-1}\left( (-1,1)^n \times \fT_x \right)$, with   $\oU_x \subset B_{\fM}(x, \eM)$.
\item  The plaques of $\vp_x$ are   strongly convex for the metric $\dF$ with diameter less than $\lF$. 
\end{enumerate}
\end{prop}


By a standard argument, there exists a finite collection  $\{x_1, \ldots , x_{\nu}\} \subset \fM$ where $\vp_{x_i}(x_i) = (0 , w_{x_i})$ for $w_{x_i} \in \fX$,  and regular foliation charts $\vp_{x_i} \colon \oU_{x_i} \to [-1,1]^n \times \fT_{x_i}$
satisfying the conditions of Proposition~\ref{prop-regular},   which form an open covering of $\fM$. Relabel the various maps and spaces accordingly, so that
$\oU_{i} = \oU_{x_i}$ and $\vp_{i} = \vp_{x_i}$. Accordingly, label the transverse spaces $\fT_i = \fT_{x_i}$ and the projection maps $\pi_i = \pi_{x_i} \colon \oU_i \to \fX_i$.
  Then the image $\pi_i(U_i \cap U_j) = \fT_{i,j} \subset \fT_i$ is a clopen subset for all $1 \leq i, j \leq \nu$.   

We also then have the transversal mappings 
$\tau_i \colon \fT_i \to \cT_i \subset \fM$ for $1 \leq i \leq \nu$.

A \emph{regular covering} of $\fM$ is a finite covering $\cU = \{U_{1}, \ldots , U_{\nu}\}$ such that for each $1 \leq i \leq \nu$ there is a foliated coordinate map 
$\ds \vp_i \colon  U_i \to (-1,1)^n \times \fT_i$  which satisfies  the   regularity  conditions in Proposition~\ref{prop-regular}.
  We assume in the following that a   regular foliated covering  of $\fM$   has been chosen.


\subsection{More metric estimates}\label{subsec-moremetric}
 We introduce lower and upper bounds on the diameters of balls in the leaves of $\F$ with respect to the ambient metric $\dM$ on $\fM$. To assist with the notation,  we use the convention that $\lambda > 0$ will  denote a small leafwise distance, and $\e$ will denote a small distance in $\fM$. Later when we introduce the target manifold $M$, we let  $\delta$   denote a small distance in $M$. 
 
 For $x \in \fM$ and $\e > 0$,   let $D_{\F}(\dM, x, \e) \subset L_x$ denote the connected component containing $x$ of the intersection $L_x \cap D_{\fM}(x, \e)$.  
Define the   continuous functions
\begin{eqnarray}
\rho(\dF,\dM, x,\e) & = & \max \ \{\dF(x',x) \mid x' \in D_{\F}(\dM, x, \e)\} \label{eq-rhoFM}\\
\rho(\dM,\dF, x,\lambda) & = & \max \ \{\dM(x',x) \mid x' \in D_{\F}(x, \lambda)\} \ . \label{eq-rhoMF}
\end{eqnarray}
Then for all $x \in \fM$,  $\e > 0$ and $\lambda > 0$,  we have
\begin{equation}
D_{\F}(\dM, x, \e) \subset D_{\F}(x, \rho(\dF,\dM, x,\e)) ~ {\rm and}~  D_{\F}(x, \lambda) \subset D_{\fM}(x, \rho(\dM,\dF, x,\lambda)) \ .
\end{equation}
As $\fM$ is compact, we can then define   the  increasing   functions of $\e > 0$ and $\lambda > 0$,
 \begin{eqnarray}
\rho(\dF,\dM, \e) & = & \max \ \{\rho(\dF,\dM, x,\e) \mid x \in \fM\} \\
\rho(\dM,\dF, \lambda) & = & \max  \ \{\rho(\dM,\dF, x,\lambda) \mid x \in \fM\} \ .
\end{eqnarray}
Moreover, for $0 < \lambda' < \lambda \leq \lF$ we have the strict inclusion $B_{\F}(x,\lambda') \subset B_{\F}(x,\lambda)$, and thus  
the function $\rho(\dM,\dF, \lambda)$ is strictly increasing for $0 < \lambda <   \lF$. 

 
Let $\e_{\F}^* = \max \ \{ \e \mid \rho(\dF, \dM, \e) \leq \lF \}$, so that $D_{\F}(\dM, x, \e_{\F}^*) \subset D_{\F}(x, \lF)$ for all $x \in \fM$.  


Introduce  the continuous function  
$\lambda_{\F}(\e)$   which is the inverse of $\rho(\dM,\dF, \lambda)$, for  $0 < \lambda \leq \lF$. Thus, $\lambda_{\F}(\e)$ is the largest radius $\lambda \leq \lF$ such that the   disk  $D_{\F}(x, \lambda)$ in the leafwise metric is contained in the ball $B_{\fM}(x,\e)$ for all $x \in \fM$. Combining the above definitions, we obtain that    for   all $x \in \fM$, 
\begin{equation}\label{eq-metriccont}
  D_{\F}(x, \lambda_{\F}(\e))   \subset  D_{\F}(\dM, x, \e) \subset  D_{\F}(x, \lF) \cap  D_{\fM}(x, \e)  \ .
\end{equation}
  
Choose $\e_0 > 0 $ so that  $\rho(\dF,\dM, \e_0) \leq \lF/2$,  and set $\lambda_0 = \lambda_{\F}(\e_0)$.
Then by the definition of $\rho(\dF,\dM, \e_0)$ and the inclusions         \eqref{eq-metriccont},   for all $x \in \fM$ we have the inclusions
\begin{equation}
D_{\F}(x, \lambda_0 )   \subset     D_{\F}(\dM, x, \e_0) \subset  D_{\F}(x, \lF/2)  \cap    B_{\fM}(x, \e_0) \ .
\end{equation}

 
Next, choose $\e_1 > 0$ so that  $\rho( \dF, \dM, \e_1) \leq \lambda_0/10$, and let $\lambda_1 = \lF(\e_1)$.
Then  for all $x \in \fM$,  
\begin{equation}\label{eq-disklambda0}
D_{\F}(x, \lambda_1)   \subset  D_{\F}(\dM, x, \e_1)  \subset D_{\F}(x, \lambda_0/10)  \cap    B_{\fM}(x, \e_1) \ .
\end{equation}
 This choice of $\e_1$ will be recalled in Section~\ref{sec-pathlifting} and Appendix~\ref{appendix-A}.
 
  
   A matchbox manifold $\fM$ is \emph{minimal} if every leaf of $\F$ is dense.

   
 
\section{Holonomy} \label{sec-holonomy}

The holonomy pseudogroup of a smooth foliated manifold $(M, \F)$ generalizes the induced dynamical systems  associated to a section of a flow. The holonomy pseudogroup for a matchbox manifold $(\fM, \F)$ is defined analogously to the smooth case.

 
 \subsection{The foliation pseudo-star group}\label{subsec-psg}
  Let $\cU = \{ \vp_{i}  \colon \oU_i \to [-1,1]^n \times \fT_i  \mid 1 \leq i \leq \nu\}$ be a regular covering of   $\fM$ as in Section~\ref{subsec-rc}. 
   A pair of indices $(i,j)$, $1 \leq i,j \leq \nu$, is said to be \emph{admissible} if the \emph{open} coordinate charts satisfy $U_i \cap U_j \ne \emptyset$.
For $(i,j)$ admissible, define clopen subsets $\fD_{i,j} = \pi_i(U_i \cap U_j) \subset \fT_i \subset \fX$.  The convexity of foliation charts imply that plaques are either disjoint, or have connected intersection. This implies that there is a well-defined homeomorphism $h_{j,i} \colon \fD_{i,j} \to \fD_{j,i}$ with domain $\fD(h_{j,i}) = \fD_{i,j}$ and range $R(h_{j,i}) = \fD_{j,i}$.
% Note that the map $h_{j,i}$ admits a continuous extension to a local homeomorphism on the closure of its domain, $\overline{h}_{j,i} \colon \overline{\fD_{i,j}} \to \overline{\fD_{j,i}}$.



The maps $\cGF^{(1)} = \{h_{j,i} \mid (i,j) ~{\rm admissible}\}$ are the transverse change of coordinates defined by the foliation charts. By definition they satisfy $h_{i,i} = Id$, $h_{i,j}^{-1} = h_{j,i}$, and if $U_i \cap U_j\cap U_k \ne \emptyset$ then $h_{k,j} \circ h_{j,i} = h_{k,i}$ on their common domain of definition. The \emph{holonomy pseudogroup} $\cGF$ of $\F$ is the topological pseudogroup modeled on $\fX$ generated by   the elements of $\cGF^{(1)}$. The elements of $\cGF$ have a standard description in terms of the ``holonomy along paths'', which we next describe.

A sequence $\cI = (i_0, i_1, \ldots , i_{\alpha})$ is \emph{admissible}, if each pair $(i_{\ell -1}, i_{\ell})$ is admissible for $1 \leq \ell \leq \alpha$, and the composition
\begin{equation}\label{eq-defholo}
h_{\cI} = h_{i_{\alpha}, i_{\alpha-1}} \circ \cdots \circ h_{i_1, i_0}
\end{equation}
has non-empty domain.
The domain $\fD_{\cI}$ of $h_{\cI}$ is the \emph{maximal clopen subset} of $\fD_{i_0} \subset \fT_{i_0}$ for which the compositions are defined.

 
 For the study of the dynamical properties of $\F$, it is necessary  to introduce   the collection of maps $\cGF^* \subset \cGF$, defined as follows.
Given any open subset $U \subset \fD_{\cI}$ we obtain a new element $h_{\cI} | U \in \cGF$ by restriction. Then set
\begin{equation}\label{eq-restrictedgroupoid}
\cGF^* = \left\{ h_{\cI} | U \mid \cI ~ {\rm admissible} ~ \& ~ U \subset \fD_{\cI} \right\} \subset \cGF ~ .
\end{equation}
  That is,  $\cGF^*$   consists of all possible restrictions of homeomorphisms of the form \eqref{eq-defholo} to open subsets of their domains.  
 However,   in the definition of    $\cGF^*$   one does not allow arbitrary unions of local homeomorphisms, unless such homeomorphisms can be obtained by restrictions to open subsets of maximal domains of words in the elements in $\cG_0$. 
  The collection of maps $\cGF^*$    is closed under the operations of compositions, taking inverses, and restrictions to open sets, and  is called a \emph{$\psg$} in the literature \cite{Matsumoto2010}.  
  
 For $g \in \cGF^*$ denote its domain by $\fD(g) \subset \fX$, then its  range   is the clopen set $\fR(g) = g(\fD(g))  \subset \fX$.
  
  
 
 \subsection{Admissible chains}\label{subsec-admissible}

Given an admissible sequence $\cI = (i_0, i_1, \ldots , i_{\alpha})$ and any $0 \leq \ell \leq \alpha$,     the truncated sequence $\cI_{\ell} = (i_0, i_1, \ldots , i_{\ell})$ is again admissible, and we introduce the holonomy map defined by the composition of the first $\ell$ generators appearing in $h_{\cI}$,
\begin{equation}\label{eq-pcmaps}
h_{\cI_{\ell}} = h_{i_{\ell} , i_{\ell -1}} \circ \cdots \circ h_{i_{1} , i_{0}}~.
\end{equation}
Given $w \in \fD(h_{\cI})$ we adopt the notation $w_{\ell} = h_{\cI_{\ell}}(w) \in \fT_{i_{\ell}}$. So $w_0 = w$ and
$h_{\cI}(w) = w_{\alpha}$.

 

Given $w \in \fD(h_{\cI})$, let $x_0 = \tau_{i_0}(w_0) \in L_{x_0}$. Introduce the \emph{plaque chain}
\begin{equation}\label{eq-plaquechain}
\cP_{\cI}(w) = \{\cP_{i_0}(w_0), \cP_{i_1}(w_1), \ldots , \cP_{i_{\alpha}}(w_{\alpha}) \} ~ .
\end{equation}

 
 For each $1 \leq i \leq \nu$, define    $\cT_{i} =  \vp_i^{-1}(0 , \fT_i) \subset    \oU_i \subset \fM$ which is a compact local transversal to $\F$. 
 


Intuitively, a plaque chain $\cP_{\cI}(w)$ is a sequence of successively overlapping convex ``tiles'' in $L_0$ starting at $x_0 = \tau_{i_0}(w_0)$, ending at
$x_{\alpha} = \tau_{i_{\alpha}}(w_{\alpha})$, and with each $\cP_{i_{\ell}}(w_{\ell})$ ``centered'' on the point $x_{\ell} = \tau_{i_{\ell}}(w_{\ell})$.
Recall that $\cP_{i_{\ell}}(x_{\ell}) = \cP_{i_{\ell}}(w_{\ell})$, so we also adopt the notation $\cP_{\cI}(x) \equiv \cP_{\cI}(w)$.



A \emph{leafwise path}  is a continuous map $\gamma \colon [0,1] \to \fM$ such that there is a leaf $L$ of $\F$ for which $\gamma(t) \in L$ for all $0 \leq t \leq 1$.
  In the following, we will assume that all paths are  piecewise differentiable.
  
 Let $\gamma$ be a leafwise path, and $\cI$ be an admissible sequence. For $w \in \fD(h_{\cI})$, we say that $(\cI , w)$ \emph{covers} $\gamma$,
if the domain of $\gamma$ admits   a partition $0 = s_0 < s_1 < \cdots < s_{\alpha} = 1$ such that  the plaque chain $\cP_{\cI}(w_0) = \{\cP_{i_0}(w_0), \cP_{i_1}(w_1), \ldots , \cP_{i_{\alpha}}(w_{\alpha}) \}$ satisfies
\begin{equation}\label{eq-cover}
\gamma([s_{\ell} , s_{\ell + 1}]) \subset int (\cP_{i_{\ell}}(w_{\ell}) )~ , ~ 0 \leq \ell < \alpha, \quad \& \quad \gamma(1) \in int( \cP_{i_{\alpha}}(w_{\alpha})).
\end{equation}
  The map $h_{\cI}$ is said to define the holonomy of $\F$ along the path $\gamma$,  and satisfies $h_{\cI}(w_0) = \pi_{i_{\alpha}}(\gamma(1))$.


Given two admissible sequences, $\cI = (i_0, i_1, \ldots, i_{\alpha})$ and
$\cJ = (j_0, j_1, \ldots, j_{\beta})$, such that both $(\cI, w_0)$ and $(\cJ, v_0)$ cover the leafwise path $\gamma \colon [0,1] \to \fM$, then
$$\gamma(0) \in int( \cP_{i_0}(w_0)) \cap int( \cP_{j_0}(v_0)) \quad , \quad \gamma(1) \in int(\cP_{i_{\alpha}}(w_{\alpha})) \cap int( \cP_{j_{\beta}}(v_{\beta}) )$$ Thus both $(i_0 , j_0)$ and $(i_{\alpha} , j_{\beta})$ are admissible, and
$v_0 = h_{j_{0} , i_{0}}(w_0)$, $w_{\alpha} = h_{i_{\alpha} , j_{\beta}}(v_{\beta})$.

The proof of the following standard observation can be found in \cite{ClarkHurder2013}.
\begin{prop}\label{prop-copc}\cite{ClarkHurder2013}
The maps $h_{\cI}$ and
$\ds h_{i_{\alpha} , j_{\beta}} \circ h_{\cJ} \circ h_{j_{0} , i_{0}}$
agree on their common domains.
\end{prop}

  

 
Two leafwise paths $\gamma , \gamma' \colon [0,1] \to \fM$ are \emph{homotopic} if there exists a family of leafwise paths $\gamma_s \colon [0,1] \to \fM$ with $\gamma_0 = \gamma$ and $\gamma_1 = \gamma'$. We are most interested in the special case when $\gamma(0) = \gamma'(0) = x$ and $\gamma(1) = \gamma'(1) = y$. Then $\gamma$ and $\gamma'$ are \emph{homotopic relative endpoints}, or \emph{endpoint-homotopic}, 
if they are homotopic with $\gamma_s(0) = x$ for all $0 \leq s \leq 1$, and similarly
$\gamma_s(1) = y$ for all $0 \leq s \leq 1$. Thus, the family of curves $\{ \gamma_s(t) \mid 0 \leq s \leq 1\}$ are all contained in a common leaf $L_{x}$. We then have the following result.

\begin{lemma}\cite{ClarkHurder2013} \label{lem-homotopic}
Let $\gamma, \gamma' \colon [0,1] \to \fM$ be endpoint-homotopic leafwise paths. Then their holonomy maps $h_{\gamma}$ and $h_{\gamma'}$ agree on some open subset $U \subset \fD(h_{\gamma}) \cap \fD(h_{\gamma'}) \subset \fT_*$. 
\end{lemma}

 

 
 
Finally, we   recall the     definition of an  {equicontinuous} pseudogroup.
\begin{defn} \label{def-equicontinuous}
The action of the    pseudogroup $\cGF$ on $\fX$ is  \emph{equicontinuous} if for all $\epsilon > 0$, there exists $\delta > 0$ such that for all $g \in \cGF^*$, if $w, w' \in \fD(g)$ and $\dX(w,w') < \delta$, then $\dX(g(w), g(w')) < \epsilon$.
Thus, $\cGF^*$ is equicontinuous as a family of local group actions.
\end{defn}
 Further   properties of the pseudogroup $\cGF$ for a matchbox manifold are discussed in  \cite{ClarkHurder2013,CHL2014,Hurder2014}.



 
 
\section{Equicontinuous holonomy} \label{sec-pathlifting}
 
In this section, we give the proof of Theorem~\ref{thm-main1}. A key point in the proof  is based on   the path lifting property for $\e$-maps between  a matchbox manifold $\fM$  and a target manifold $M$. 
The philosophy of path lifting is   folklore, as observed by Bob Edwards in his 1978  ICM address on $\e$-approximations and homeomorphisms \cite[Section~4]{Edwards1980}. 
We develop  this technique in the context of  matchbox manifolds, making use of   standards results for $\e$-maps along  with   properties of the holonomy maps.
  
 Let $\fM$ be a   manifold-like matchbox manifold.     Let $\cU = \{ \vp_{i}  \colon \oU_i \to [-1,1]^n \times \fT_i  \mid 1 \leq i \leq \nu\}$ be a regular covering of   $\fM$ as in Section~\ref{subsec-rc}.
 Let $\cGF^*$ be the $\psg$ associated to the regular covering $\cU$ as in    Section~\ref{subsec-psg}.
We must show that the conditions of Definition~\ref{def-equicontinuous} are satisfied: 
  given $\epsilon > 0$, we must show there exists $\delta > 0$ such that  for each admissible chain $\cI$ with holonomy $h_{\cI}$, if $w, w' \in \fD(h_{\cI}) \subset \fX$ and $\dX(w,w') < \delta$, then $\dX(h_{\cI}(w), h_{\cI}(w')) < \e$.
  
  
Recall that $\eU > 0$ denotes a Lebesgue number for the covering $\cU$ of $\fM$. That is, for each $x \in \fM$, there exists some index $1 \leq i \leq \nu$ such that $B_{\fM}(x, \eU) \subset U_{i}$. 
 
     
Since  $\oU_i$   is compact for each $1 \leq i \leq \nu$, there exists a   uniform modulus of continuity function   $\rho_{\cU}(\e) > 0$ for the projections $\pi_i$:  let $\e > 0$, then   $\rho_{\cU}(\e)$ is the largest value  such that 
\begin{equation}\label{eq-unifcontU}
B_{\fM}(x,\rho_{\cU}(\e)) \cap \oU_i \subset \pi_i^{-1}(B_{\fX_i}(\pi_i(x), \e)) ~ {\rm for ~ all}~  x \in  \oU_i \ . 
\end{equation}
For $\e_1$ as defined in Section~\ref{subsec-metric}, define $\eF$ by 
\begin{equation}\label{eq-eF}
0 < \eF = \min \ \{\rho_{\cU}(\e), \e_1/2 ,  \eU/4  \} \ .
\end{equation}

  
   

   
 \subsection{Continuity estimates}\label{subsec-contest}
 
 
Choose an $\eF$-map $f \colon \fM \to M$ onto the compact topological manifold $M$, where for simplicity we omit the subscript $\eF$ in the notation for $f$.
Let $d_M$ be a metric on $M$. 
For   $w \in M$ and $\delta > 0$, let   $B_{M}(w, \delta) =\{w' \in M \mid d_M(w',w) < \delta\}$  denote the open disk in $M$ of radius $\delta$, and     $D_{M}(w, \delta) =\{w' \in M \mid d_M(w',w) \leq \delta\}$   denote the closed disk in $M$ of radius $\delta$. 


 Assume that $d_M$ is chosen so that there exists 
a constant    $\delta_M >0$  such that for all $w \in M$ and $0 < \delta \leq \delta_M$, the   disk $D_{M}(w, \delta)$ is homeomorphic to a disk in $\mR^n$. For example, if $M$ is a smooth Riemannian manifold, then   let $\delta_M > 0$ be   such that each   disk  $D_{M}(w, \delta_M)$ is strongly convex. 
 
   
  

Since $\fM$ is compact, there exists a   uniform modulus of continuity function   $\e_f(\delta) > 0$ for $f$:  for $\delta > 0$, the constant $\e_f(\delta)$ is the largest value  such that 
\begin{equation}\label{eq-unifcont}
B_{\fM}(x,\e_f(\delta)) \subset f^{-1}(B_{M}(f(x), \delta)) ~ {\rm for ~ all}~  x \in \fM\ . 
\end{equation}
Let     $\delta_f^* > 0$ be the largest radius such that $\e_f(\delta) \leq \e_{\F}$ for all $0 < \delta \leq \delta_f^*$. 

 
  Set $\lambda_{\F,f}(\delta) = \lambda_{\F}(\e_f(\delta))$, which is
well-defined for $0 < \delta \leq \delta_f^*$. Recall from Section
\ref{subsec-moremetric} that $\lambda_{\F,f}(\delta)$ is then �  the
largest radius $\lambda \leq \lF$ such that the ��disk �$D_{\F}(x,
\lambda)$ in the leafwise metric is contained in the ball
$B_{\fM}(x,\e_f(\delta))$ for all $x \in \fM$.

 Combining \eqref{eq-metriccont} and \eqref{eq-unifcont} we obtain a leafwise modulus of continuity for $f$:  
 \begin{equation}\label{eq-unifleafcont}
  D_{\F}(x, \lambda_{\F,f}(\delta) )   \subset  B_{\fM}(x,\e_f(\delta)) \subset f^{-1}(B_{M}(f(x), \delta)) \quad {\rm for ~ all}~ 0 < \delta \leq \delta_f^* ~, ~ x \in \fM\ .   
\end{equation}
 As $f$ is an $\eF$-map, we have that  $f^{-1}(f(x)) \subset B_{\fM}(x,\eF)$ for all $x \in \fM$. 
As $M$ is compact and $f^{-1}(f(x))$ is a compact set with diameter at most $\eF$ for all $x \in \fM$,  there exists  $0 < \delta_1   \leq \delta_M/10$ so  that  for all $x \in \fM$, we have 
 \begin{equation}\label{eq-diskdelta1}
 f^{-1}(D_M(f(x),\delta_1)) \subset B_{\fM}(x,2\eF) = B_{\fM}(x,\e_1) \ .
 \end{equation} 
 
  
Let $\lambda_2 =  \lambda_{\F,f}(\delta_1)$; that is, $\lambda_2$ is then �is the largest radius $\lambda \leq \lF$
such that the ��disk �$D_{\F}(x, \lambda)$ in the leafwise metric is
contained in the ball $B_{\fM}(x,\delta_1)$ for all $x \in \fM$.

Then for all $x \in \fM$,  by \eqref{eq-unifleafcont},  \eqref{eq-disklambda0},  and the choice of $\delta_1$ we have
  \begin{equation}\label{eq-diameter}
D_{\F}(x, \lambda_2 )   \subset f^{-1}(D_M(f(x),\delta_1)) \cap D_{\F}(x,\lF)   \subset D_{\fM}(x,\e_1) \cap D_{\F}(x,\lF)   \subset D_{\F}(x,\lambda_0/10) \ .
\end{equation}

 Finally, we     require a basic result  concerning $\e$-maps on matchbox manifolds, which is a type of converse to the inclusions in \eqref{eq-diameter}, and whose   proof  is   in the spirit of the work by Eilenberg  \cite{Eilenberg1938}. The proof of the following  is deferred to  Appendix~\ref{appendix-A}.



\begin{prop}\label{prop-coverage}
Let $\fM$ be a matchbox manifold with leafwise Riemannian metric on $\F$. Then 
there exists $\eF > 0$  such that,    if $f \colon \fM \to M$ is an $\eF$-map to a compact manifold $M$, then for      $x_0 \in \fM$ with $w_0 = f(x_0)$,  we have  $B_M(w_0,\delta_1) \subset f(D_{\F}(x_0, \lF/2 ))$.
\end{prop}

  

   \subsection{Local lifting property}\label{subsec-extensions}

 
   
  
  We next establish  a   technical result used in the proof of Theorem~\ref{thm-main1}. 
 For $\delta_1$ as chosen above so that \eqref{eq-diskdelta1} holds, set $\e_1' = \e_f(\delta_1)$.  
  

 \begin{lemma}\label{lem-locallifting}
Let $f \colon \fM \to M$ be an $\eF$-map. Let  $x_0 \in \fM$ and suppose that  
$B_{\fM}(x_0,\eU) \subset U_i$ for some   $1 \leq i \leq \nu$. Then for  $z \in B_{\fM}(x_0,\e_1')$ and $y \in D_{\F}(x , \lambda_2) \subset \cP_i(x)$, we have $\ds   f^{-1}(f(y)) \cap \cP_i(z) \ne \emptyset$.
\end{lemma}
 \proof
 
Set $w_0 = f(x_0) \in M$, then  by \eqref{eq-unifcont} we have that   
  $B_{\fM}(x_0,\e_1') \subset f^{-1}(B_{M}(w_0, \delta_1)) \subset U_i$ so that  $f(z) \in B_{M}(w_0, \delta_1)$.
  Moreover,  by Proposition~\ref{prop-coverage}, we have that $B_{M}(w_0, \delta_1) \subset  f(D_{\F}(x_0, \lF/2 ))$.  
  Then choose $x_z \in D_{\F}(x_0, \lF/2 )$ with $f(x_z) = f(z)$.
     
Let  $\gamma_y \colon [0,1] \to \cP_i(x_0)$   be the geodesic  path in  $\cP_i(x_0)$ with  $\gamma_y(0) = x_z$ and $\gamma_y(1) = y$. 

  Let $0 \leq s_* \leq 1$ be the largest value such that 
 $$ f^{-1}(f((\gamma_y(s))) \cap \cP_i(z) \ne \emptyset \quad {\rm for ~ all} \quad  0 \leq s \leq s_* \ .$$ 
 We claim that $s_* =1$. 
Suppose that $s_* < 1$, then we show this yields a contradiction.

Set  $x_* =  \gamma_y(s_*)$ and $w_* = f(x_*)$. Then  there exists  $z_* \in  f^{-1}(w_*) \cap \cP_i(z)$ by the definition of $s_*$.
Note that  $x_* \in  B_{\fM}(x_0,\eF) \subset   B_{\fM}(x_0,\eU/4)$ by the choice of $\lambda_2$ and the fact that $D_{\F}(x_0 , \lambda_2)$ is strongly  convex.
By the choice of $f$ and $\eF$ in \eqref{eq-eF},  we have $\dM(x_* , z_*) < \eF < \eU/4$.
Thus  $\dM(x_0 , z_*) < \eU/2$ and hence $B_{\fM}(z_*, \eF) \subset B_{\fM}(x_0 ,\eU) \subset U_i$.
It then follows from the choice of $\lambda_2$ and the above observations that  
 \begin{equation}\label{eq-disk0}
D_{\F}(z_* , \lambda_2) \subset  B_{\fM}(z_*,\eF)  \subset U_i \ .
\end{equation}


The value of $\eF > 0$ is less than or equal to the choice $\e_1/2$ for this constant in the proof of   Proposition~\ref{prop-coverage} in Appendix~\ref{appendix-A}, and thus 
$B_M(w_*,\delta_1)  \subset f(D_{\F}(z_* , \lF/2))$.  
  The assumption that $s_* <1$ implies that for $s_* \leq s < 1$ sufficiently small so that  $f(\gamma_z(s)) \in B_M(w_*,\delta_1)$, we have that 
    $$  \cP_i(z) \cap f^{-1}(f(\gamma_z(s))) = D_{\F}(z_* , \lF/2) \cap f^{-1}(f(\gamma_z(s))) \ne \emptyset$$
  which contradicts the choice of $s_*$. 
   \endproof
  
 
   
  We next extend the conclusion of Lemma~\ref{lem-locallifting} from paths contained in a coordinate chart, to leafwise paths defined by a plaque chain of arbitrary length. 

   
 
  Let $\cI = (i_0, i_1, \ldots, i_{\alpha})$ be an admissible chain   with associated holonomy map $h_{\cI} \in \cGF^*$ and $w_0 \in \fD(h_{\cI})$. 
  As in    Section~\ref{subsec-admissible}, we associate to the pair $(\cI, w_0)$ the plaque chain 
$\cP_{\cI}(w_0) = \{\cP_{i_0}(w_0), \cP_{i_1}(w_1), \ldots , \cP_{i_{\alpha}}(w_{\alpha}) \}$ given in \eqref {eq-plaquechain}.

Next   introduce a plaque chain $\cJ = (j_0, j_1, \ldots, j_{\beta})$ which is a refinement of $\cI$ at $w_0$ and is chosen with respect to the leaf distance constant  $\lambda_2 > 0$ which was defined so that the inclusions in \eqref{eq-diameter} hold. 
By   Proposition~\ref{prop-copc}, its associated holonomy map $h_{\cJ}$ at $w_0$ agrees with the holonomy map   $h_{\cI}$ at $w_0$ on their common domains.

  
Let   $\gamma \colon [0,\alpha] \to L_{x_0} \subset \fM$ be the leafwise piecewise geodesic  associated to the plaque chain $\cP_{\cI}(x_0)$. That is, $\gamma \colon [0,\alpha] \to L_{x_0}$ is the concatenation of geodesic segments $\{\gamma_{\ell} \mid 0 \leq \ell \leq \alpha - 1\}$ in the plaques of the covering $\cU$, where 
$\gamma_{\ell} \colon [\ell  , \ell +1] \to \cP_{i_{\ell}}(w_{\ell})$ satisfies 
$$\gamma_{\ell}(\ell) = x_{\ell} = \tau_{i_{\ell}}(w_{\ell}) \in \cT_{i_{\ell}} \quad , \quad \gamma_{\ell}(\ell +1) = x_{\ell +1} = \tau_{i_{\ell +1}}(w_{i_{\ell +1}})  \in \cT_{i_{\ell +1}} \ .$$
% In particular,    $\gamma(0) = x_0  \in \cT_{i_{0}}$ and $\gamma(\ell) = x_{\alpha} \in \cT_{i_{\alpha}}$.


   
 Introduce a subdivision of the interval $[0,\alpha]$, given by $0 = s_0 <
s_1 < s_2 < \cdots < s_{\beta} = \alpha$, where there is an increasing
subsequence $\{\ell \mid 0 \leq \ell \leq \alpha = s_\beta \}$. For
notational convenience, set $s_{-1} = s_0 = 0$ and $s_{\beta+1} =
s_{\beta} = \alpha$.
Then set $\xi_{\ell} = \gamma(s_{\ell})$ for $-1 \leq \ell \leq \beta+1$, and we
choose the subdivision so ���that �for each $0 \leq \ell \leq \beta$,
\begin{equation}\label{eq-pathdistances}
\dF(\gamma(s), \xi_{\ell}) < \lambda_2 ~{\rm for}~ s_{\ell-1} \leq s \leq s_{\ell+1} \ .
\end{equation}
 
 
  For each $0 \leq \ell \leq \beta$, choose an index $1 \leq j_{\ell} \leq \nu$ so that $ B_{\fM}(\xi_{\ell}, \eU) \subset U_{j_{\ell}}$. 
   It then follows by the choice of $\eF$, $\lambda_2$ and \eqref{eq-diameter} that  for each $0 \leq \ell \leq \beta$, we have 
\begin{equation}\label{eq-disk2}
\gamma(s) \in  D_{\F}(\xi_{\ell} , \lambda_2) \subset  B_{\fM}(\xi_{\ell},\eF) \subset B_{\fM}(\xi_{\ell}, \eU/4) \quad {\rm for ~ all} ~ s_{\ell-1} \leq s \leq s_{\ell+1} \ .
\end{equation}
Moreover, $\dF(\xi_{\ell}, \xi_{\ell +1}) < \lambda_2$ implies that  
\begin{equation}\label{eq-palquechain2}
\xi_{\ell+1} \in   D_{\F}(\xi_{\ell} , \lambda_2) \cap D_{\F}(\xi_{\ell +1} , \lambda_2) \subset \cP_{j_{\ell}}(\xi_{\ell}) \cap \cP_{j_{\ell +1}}(\xi_{\ell +1}) \ .
\end{equation}
 
Thus     $\cJ  = (j_0, j_1, \ldots, j_{\beta})$ is an admissible sequence, and  
 $\ds \cP_{\cJ}(w_0) = \{\cP_{j_0}(\xi_0), \cP_{j_1}(\xi_1), \ldots , \cP_{j_{\beta}}(\xi_{\beta}) \}$ defines 
  a holonomy map   $h_{\cJ}$    at $w_0$.
 
 Now let $\eF > 0$ be as above,   $\xi_0 = x_0  \in \cT_{j_0}$ for the plaque chain $\cJ$ as chosen above, and 
 suppose that  $\cP_{j_0}(z_0)  \cap f^{-1}(f(\xi_0)) \ne \emptyset$ for some $z_0 \in \fT_{j_0}$.   
 Then $\dF(\xi_0, \xi_1) < \lambda_2$  by \eqref{eq-pathdistances}, so by \eqref{eq-palquechain2} we have 
$\ds \xi_1 \in  \cP_{j_0}(\xi_0) \cap \cP_{j_1}(\xi_1)$. 
Hence by  Lemma~\ref{lem-locallifting} there exists  $\ds  z_1' \in  f^{-1}(f(\xi_1)) \cap \cP_{j_0}(z_0)$.

Note that      $\dM(\xi_1, z_1') \leq \eF \leq \eU/4$, so $z_1' \in B_{\fM}(\xi_1,\eU/4) \subset U_{j_1}$.
  Thus, there exists   $z_1 \in  \cT_{j_1}$   such that $z_1' \in \cP_{j_1}(z_1)$ and hence $\cP_{j_0}(z_0) \cap    \cP_{j_1}(z_1) \ne \emptyset$.

We now repeat the application of  Lemma~\ref{lem-locallifting} to the new basepoint $\xi_1$, and then continue recursively to obtain a sequence of points 
$\ds \{ z_{\ell} \in   \cT_{j_{\ell}}  \mid 0 \leq \ell \leq \beta \}$ such that for $0 < \ell \leq \beta$ we have:
\begin{itemize}
 \item $\cP_{j_{\ell-1}}(z_{\ell-1}) \cap    \cP_{j_{\ell}}(z_{\ell}) \ne \emptyset$ , 
\item $\ds  z_{\ell}' \in  f^{-1}(f(\xi_{\ell})) \cap \cP_{j_{\ell}}(z_{\ell})$.
\end{itemize}


Recall that $\pi_{i} \colon \oU_{i} \to \fT_{i}$ for $1 \leq i \leq \nu$ is the transverse projection to the model space $\fT_i$.
Then the above shows that for $w_0 = \pi_{j_0}(z_0)$ and $w_{\beta} = \pi_{j_{\beta}}(z_{\beta})$ we have $w_0 \in   \fD(h_{\cJ})$ and $h_{\cJ}(w_0) = w_{\beta}$. 

  \subsection{Proof of Theorem~\ref{thm-main1}}\label{subsec-proof}
We can now complete the proof of Theorem~\ref{thm-main1}.  We have assumed that $\e > 0$ is given, and $\eF > 0$ is defined as in \eqref{eq-eF}. 
Then choose an $\eF$-map $f$ as in Section~\ref{subsec-contest}.
Let $h_{\cI} \in \cGF^*$ be as in Section~\ref{subsec-extensions}, and $\cP_{\cJ}$ the path chain constructed above from $\cI$. 

   
 
 
For each $1 \leq i \leq \nu$ the transversal map $\tau_i \colon \fT_i \to \cT_i$ is a homeomorphism of compact spaces, 
and the metric $\dTi$ on the subspace $\cT_i \subset \oU_i \subset \fM$ was defined in Section~\ref{subsec-fs}  as the restriction of   $\dM$.

Recall that $\e_1' = \e_f(\delta_1)$ was defined in Section~\ref{subsec-extensions} and used in the hypothesis of Lemma~\ref{lem-locallifting}. 
By the uniform continuity of the maps $\tau_i$, there exists  $\delta > 0$   such that for all $1 \leq i \leq \nu$ and $w \in \fT_i$,
\begin{equation}
B_{\fT_i}(w, \delta) \subset \tau_i^{-1}(B_{\cT_i}(\tau_i(w), \e_1' ))  \ .
\end{equation}
It thus follows from the above results that $h_{\cJ}(B_{\fT_{j_0}}(w, \delta)) \subset B_{\fT_{j_{\beta}}}(h_{\cJ}(w), \e)$, as was to be shown. 

  

 


 
\appendix

\section{Local surjectivity for $\e$-maps}\label{appendix-A}



In this appendix, we give a technical result  concerning $\e$-maps.  


\begin{prop}\label{Aprop-coverage}
Let $\fM$ be a matchbox manifold with leafwise Riemannian metric on $\F$. Then 
there exists $\eF > 0$  such that,    if $f \colon \fM \to M$ is an $\eF$-map to a compact manifold $M$, then there exists $\delta_1 > 0$ such that for      $x_0 \in \fM$ with $w_0 = f(x_0)$,  we have $D_M(f(x_0),\delta_1) \subset f(D_{\F}(x_0, \lF/2 ))$.
\end{prop}
\proof
   We use the notations of Section~\ref{subsec-metric} above.
   
Choose $\e_0 > 0 $ so that �$\rho(\dF,\dM, \e_0) \leq \lF/2$, as
defined by \eqref{eq-rhoFM} to be the maximal radius of a leafwise disk about $x$ 
contained in a closed disk of radius $\e_0$ in $\fM$, for \emph{all} $x \in \fM$. 

Set $\lambda_0 = \lambda_{\F}(\e_0)$, which is
defined in Section \ref{subsec-moremetric} to be the largest
radius $\lambda \leq \lF$ such that the ��disk �$D_{\F}(x, \lambda)$ is
contained in $B_{\fM}(x,\e_0)$, for \emph{all} $x \in \fM$.
Then by the definition of $\rho(\dF,\dM, \e_0)$ and the inclusions         \eqref{eq-metriccont},   for all $x \in \fM$ we have the inclusions
\begin{equation}
D_{\F}(x, \lambda_0 )   \subset     D_{\F}(\dM, x, \e_0) \subset  D_{\F}(x, \lF/2)  \cap    B_{\fM}(x, \e_0) \ .
\end{equation}

 
Next, choose $\e_1 > 0$ so that  $\rho( \dF, \dM, \e_1) \leq \lambda_0/10$, and let $\lambda_1 = \lF(\e_1)$.
Then  for all $x \in \fM$,  
\begin{equation}\label{Aeq-disklambda0}
D_{\F}(x, \lambda_1)   \subset  D_{\F}(\dM, x, \e_1)  \subset D_{\F}(x, \lambda_0/10)  \cap    B_{\fM}(x, \e_1) \ .
\end{equation}

   
 Set $ \eF = \e_1/2$. This constant is chosen so that  the result \cite[Section~1, Th\'eor\`eme]{Eilenberg1938} by Eilenberg holds uniformly for strongly convex   compact subsets 
of  $D_{\F}(x, \lF) \subset L_x$, as will be shown below.

 

 Let $f \colon \fM \to M$ be  an $\eF$-map, which is   onto the compact manifold $M$.  
Let $d_M$ be a metric on $M$. 
For   $w \in M$ and $\delta > 0$, let   $B_{M}(w, \delta) =\{w' \in M \mid d_M(w',w) < \delta\}$  denote the open disk in $M$ of radius $\delta$, and     $D_{M}(w, \delta) =\{w' \in M \mid d_M(w',w) \leq \delta\}$   denote the closed disk in $M$ of radius $\delta$. 

Assume that $d_M$ is chosen so that there exists 
a constant    $\delta_M >0$  such that for all $w \in M$ and $0 < \delta \leq \delta_M$, the   disk $D_{M}(w, \delta)$ is homeomorphic to a disk in $\mR^n$. For example, if $M$ is a Riemannian manifold, then   let $\delta_M > 0$ be   such that each   disk  $D_{M}(w, \delta_M)$ is strongly convex. 

Since $\fM$ is compact, there exists a   uniform modulus of continuity function   $\e_f(\delta) > 0$ for $f$:  for $\delta > 0$, the constant $\e_f(\delta)$ is the largest value  such that 
\begin{equation}\label{Aeq-unifcont}
B_{\fM}(x,\e_f(\delta)) \subset f^{-1}(B_{M}(f(x), \delta)) ~ {\rm for ~ all}~  x \in \fM\ . 
\end{equation}
Let     $\delta_f^* > 0$ be the largest radius such that $\e_f(\delta) \leq \e_{\F}$ for all $0 < \delta \leq \delta_f^*$. 

 
Set $\lambda_{\F,f}(\delta) = \lambda_{\F}(\e_f(\delta))$, which is well-defined for $0 < \delta \leq \delta_f^*$. 
 Combining \eqref{eq-metriccont} and \eqref{Aeq-unifcont} we obtain a leafwise modulus of continuity for $f$:  
 \begin{equation}\label{Aeq-unifleafcont}
  D_{\F}(x, \lambda_{\F,f}(\delta) )   \subset  B_{\fM}(x,\e_f(\delta)) \subset f^{-1}(B_{M}(f(x), \delta)) \quad {\rm for ~ all}~ 0 < \delta \leq \delta_f^* ~, ~ x \in \fM\ .   
\end{equation}


 
  
 As $f$ is an $\eF$-map, we have that  $f^{-1}(f(x)) \subset B_{\fM}(x,\eF)$ for all $x \in \fM$. 
As $M$ is compact and $f^{-1}(f(x))$ is a compact set with diameter at most $\eF$ for all $x \in \fM$,  there exists  $0 < \delta_1   \leq \delta_M/10$ so  that  for all $x \in \fM$, we have 
 \begin{equation}\label{Aeq-diskdelta1}
 f^{-1}(D_M(f(x),\delta_1)) \subset B_{\fM}(x,2\eF) = B_{\fM}(x,\e_1) \ .
 \end{equation} 
 
  
Let $\lambda_2 =  \lambda_{\F,f}(\delta_1)$ so that for all $x \in \fM$,  by \eqref{Aeq-unifleafcont},  \eqref{Aeq-disklambda0},  and the choice of $\delta_1$ we have
  \begin{equation}\label{Aeq-diameter}
D_{\F}(x, \lambda_2 )   \subset f^{-1}(D_M(f(x),\delta_1)) \cap D_{\F}(x,\lF)   \subset D_{\fM}(x,\e_1) \cap D_{\F}(x,\lF)   \subset D_{\F}(x,\lambda_0/10) \ ,
\end{equation}
  \begin{equation}\label{Aeq-diameter2}
  f(D_{\F}(x, \lambda_2 )) \subset D_M(f(x), \delta_1)  \subset D_M(f(x), \delta_M/10) \ .
  \end{equation} 

 
 
 Set $\lambda_3 = 5 \lambda_2$ so that by \eqref{Aeq-diameter} we have  $\lambda_3 \leq \lF/2$. 

  
\begin{lemma}\label{lem-inclusions}
 Let $x \in \fM$ and   $w = f(x) \in M$, then  $f(D_{\F}(x ,   \lambda_3)) \subset D_M(w, \delta_M)$.
\end{lemma}
\proof
Let $y \in D_{\F}(x, \lambda_3) \subset D_{\F}(x, \lF/2)$ and let $\sigma \colon [0,1] \to D_{\F}(x,\lambda_3)$ be the unique geodesic segment with $\sigma(0) = x$ and $\sigma(1) = y$. 
Set $x_i = \sigma(i/5)$ for $0 \leq i \leq 5$, then $x_{5} = y$. Note that as $\sigma$ is a geodesic, we have 
$$\dF(x_i, x_{i+1}) = \dF(x ,y)/5 \leq \lambda_3/5 = \lambda_2 ~ , ~ 0 \leq i < 5 \ .$$
Set $w_i = f(x_i)$ for $0 \leq i \leq 5$. 

Then for each $0 \leq i \leq 5$, by \eqref{Aeq-diameter2} we have 
$\ds D_{\F}(x_i, \lambda_2) \subset f^{-1}(D_M(w_i,\delta_1))$, 
so that the collection $\{D_M(w_i,\delta_1) \mid 0 \leq i \leq 5\}$ is a covering of the image of $\sigma$. 
Thus, $d_M(w_0 , w_{5}) \leq 10 \delta_1 \leq \delta_M$, hence $d_M(w, f(y)) \leq \delta_M$,  as was to be shown.
\endproof


  

 For    $x \in \fM$ and $0 < \lambda \leq \lF$,  introduce the following leafwise sets: 
 \begin{eqnarray*}
S_{\F}(x, \lambda) & = &  \{y \in D_{\F}(x, \lF) \mid   \dF(y,x) = \lambda\}   \\
D_{\F}^*(x, \lambda) & = &  \{y \in   D_{\F}(x, \lF) \mid 0 < \dF(y,x) \leq \lambda\} \ .
\end{eqnarray*}
Then we have $S_{\F}(x, \lambda) \subset D_{\F}^*(x, \lambda) \subset D_{\F}(x, \lambda)\subset B_{\F}(x, \lF)$.

 
Now let  $x_0 \in \fM$ with $w_0 = f(x_0)$. 
 We claim that $B_M(w_0,\delta_1) \subset f(D_{\F}(x_0, \lF/2))$. 
Suppose not, then we show this yields a contraction.

 

 Let $w_2 \in B_M(w_0,\delta_1) $ but $w_2 \not\in f(D_{\F}(x_0, \lF/2))$. Then there exists $0 < \delta_2 <  \delta_1$ such that 
 \begin{equation}\label{Aeq-delta2}
B_M(w_2,\delta_2)  \subset B_M(w_0,\delta_1) \quad , \quad B_M(w_2,\delta_2) \cap f(D_{\F}(x_0, \lF/2)) = \emptyset \ .
\end{equation}
We consider the maps on \v{C}ech cohomology induced by $f$ to obtain the contradiction.

  
For $0 < \lambda <    \lF/2$, introduce the collections of open sets in $L_{x_0}$:
\begin{eqnarray}
\cL(\lambda) & = &   \{B_{\F}(x, \lambda)   \mid x \in  D_{\F}(x_0, \lF/2) \}  \\
\cL^*(\lambda) & = &  \{ B_{\F}(x, \lambda) \mid x \in  D_{\F}^*(x_0, \lF/2) ~ {\rm such ~ that} ~ B_{\F}(x, \lambda) \subset D_{\F}^*(x_0, \lF) \}    \ .
\end{eqnarray}


For each $x \in \fM$, the disk $B_{\F}(x, \lambda) \subset L_x$  is strongly convex,   thus $\cL(\lambda)$ is a good covering of $D_{\F}(x_0,\lF/2)$  in the sense of \v{C}ech theory.
Let $\|\cL(\lambda)\|$ denote the simplicial space which is the geometric realization of the collection $\cL(\lambda)$.

For $0 < \lambda < \lambda' < \lF/2$, each open disk $B_{\F}(x, \lambda) \in  \cL(\lambda)$ is contained in the disk $B_{\F}(x, \lambda') \in  \cL(\lambda')$ which induces a   map between  the realizations of their nerve complexes, $\iota \colon  \|\cL(\lambda)\| \to  \|\cL(\lambda')\|$, which is a homotopy equivalence as all the sets in the cover are strongly convex.
Similarly, the induced map on the nerve complex induces  a homotopy equivalence $\iota \colon  \|\cL^*(\lambda)\| \to  \|\cL^*(\lambda')\|$.

 
 For $0 < \delta < \delta_2$, where $\delta_2$ was chosen so that \eqref{Aeq-delta2} holds, introduce the collections of open sets in $M$:  
  \begin{eqnarray}
\cM_{\delta_2}(\delta) & = &    \{B_M(w, \delta)  \mid w \in   D_M(w_0, \delta_M) - B_M(w_2,\delta_2) \} \\
\cM_f(\delta) & = &    \{B_M(w, \delta)    \mid w \in f(D_{\F}(x_0, \lambda_3)) \} \\
\cM_{S}(\delta) & = &  \{ B_M(w, \delta)    \mid w \in  f(S_{\F}(x_0, \lambda_3) )   \}    \ .
\end{eqnarray}

Recall that by Lemma~\ref{lem-inclusions}, $f(S_{\F}(x, \lambda_3)) \subset D_M(w, \delta_M)$, so   by the choice of $\delta_2$ we have inclusions 
$\cM_{S}(\delta) \subset \cM_f(\delta) \subset \cM^*(\delta)$, and so obtain maps of their simplicial realizations
\begin{equation}\label{Aeq-compositions0}
 \|\cM_{S}(\delta)\| \longrightarrow \|\cM_f(\delta) \| \longrightarrow \|\cM_{\delta_2}(\delta)\|  \ .
\end{equation}

 
As the disk $D_{\F}(x_0, \lF)$ is strongly convex, there is a natural map $\cR_{\lambda} \colon \|\cL(\lambda)\| \to D_{\F}(x_0, \lF/2)$ which maps a simplex in   the realization $\|\cL(\lambda)\|$    to the geodesic simplex in $D_{\F}(x_0, \lF/2)$ spanned by its vertices. Then $\cR_{\lambda}$ induces   isomorphisms
\begin{eqnarray}
\cR_{\lambda}^*   \colon    \{0\} & \cong &  H^n(D_{\F}(x_0,\lF/2) ; \mZ)   \to   H^n(\|\cL(\lambda)\|; \mZ)  \label{Aeq-compositions1} \\
\cR_{\lambda}^*   \colon    \mZ & \cong &  H^{n-1}(D_{\F}^*(x_0, \lF/2) ; \mZ)   \to   H^{n-1}(\|\cL^*(\lambda)\|; \mZ) \ . \label{Aeq-compositions2}
\end{eqnarray}
   Similarly, for $0 < \delta < \delta_2$,  there is a continuous map $\cS_{\delta_2} \colon \|\cM_{\delta_2}(\delta)\| \to D_M(w, \delta_M) - B_M(w_2,\delta_2)$.
   
  Let  $0 < \delta_3 \leq \delta_2$ be sufficiently small so that we have  an inclusion map 
\begin{equation}\label{Aeq-compositions3}
\cS_{\delta_2}^* \colon \mZ \cong H^{n-1}(\{ D_M(w_0, \delta_M) - B_M(w_2,\delta_2) \} ; \mZ) \to H^{n-1}(\|\cM_{\delta_2}(\delta_3) \| ;\mZ) \ .
\end{equation}


 


We next consider the maps induced by $f$ on  the cohomology groups in   \eqref{Aeq-compositions1}, \eqref{Aeq-compositions2} and \eqref{Aeq-compositions3}. 


Let   $\lambda_4 =  \lambda_{\F,f}(\delta_3)$ so that  for all $x \in \fM$, we have  the inclusion  $ f(D_{\F}(x, \lambda_4)) \subset D_M(f(x), \delta_3)$.
 Then $f$ induces an inclusion map
 $\cU_f \colon \cL(\lambda_4) \to \cM_{\delta_2}(\delta_3)$, which induces a map of their realizations
 \begin{equation}
\| \cU_f\| \colon \| \cL(\lambda_4)\|  \to \| \cM_{\delta_2}(\delta_3) \| \ .
\end{equation}


We have that $0 < \delta_3 \leq \delta_2 <  \delta_1$ so that by  \eqref{Aeq-diskdelta1} and \eqref{Aeq-diameter}, for all $x \in \fM$  we have  
 \begin{equation}\label{Aeq-lifting}
  f^{-1}(D_M(f(x),\delta_3)) \cap D_{\F}(x,\lF)     \subset D_{\F}(x,\lambda_0/10) \ .
\end{equation}
 For each $B_M(w, \delta_3) \in \cM_f(\delta_3)$ choose $x \in f^{-1}(w) \cap D_{\F}(x_0,\lF)$, then the inclusion \eqref{Aeq-lifting} holds. Thus, 
  $f^{-1}$ induces an inclusion map
 $\cV_f \colon  \cM_{\delta_2}(\delta_3) \to \cL(\lambda_0/10)$, which in turn induces a map between their realizations
 \begin{equation}
\| \cV_f\| \colon  \| \cM_{\delta_2}(\delta_3) \| \to \| \cL(\lambda_0/10)\|  \ .
\end{equation}
  Note that the composition $\| \cV_f\| \circ \| \cU_f\| \colon \| \cL(\lambda_4)\| \to \| \cL(\lambda_0/10)\|$ is a homotopy equivalence, as   the sets in $\cL(\lambda_0/10)$ are strongly convex. 
  
  
  
Next, for $0 < \lambda \leq \lambda_0/10$,  introduce the   collection  of open balls centered at points of $S_{\F}(x_0, \lambda_3)$, 
\begin{equation}\label{Aeq-sphere}
\cL_{S}(\lambda)   =    \{ B_{\F}(x, \lambda) \mid x \in  S_{\F}(x_0, \lambda_3) ~ {\rm such ~ that} ~ B_{\F}(x, \lambda) \subset D_{\F}^*(x_0, \lF) \} \ .
\end{equation}
As above, the map $f$ induces  maps of geometric realizations 
 \begin{eqnarray}
\| \cU_f\| & \colon & \| \cL_{S}(\lambda_4)\|  \to \| \cM_{S}(\delta_3) \|   \\
\| \cV_f\| & \colon &  \| \cM_{S}(\delta_3) \| \to \| \cL_{S}(\lambda_0/10)\|  \ ,
\end{eqnarray}
and the composition $\| \cV_f\| \circ \| \cU_f\| \colon \| \cL_{S}(\lambda_4)\| \to \| \cL_{S}(\lambda_0/10)\|$ is a homotopy equivalence.

The space $\| \cL_{S}(\lambda_0/10)\|$ is the simplicial realization of a good covering of $S_{\F}(x_0, \lambda_3)$, so we have 
$\ds H^{n-1}( \| \cL_{S}(\lambda_0/10)\|, \mZ) \cong \mZ$.  Let $\omega \in H^{n-1}( \| \cL_{S}(\lambda_0/10)\|, \mZ)$ be a choice of a generator.

Note that the compact sets $f(S_{\F}(x, \lambda))$ in  $M$   limit to the point $w = f(x)$ as $\lambda \to 0$. It follows that there is a non-trivial class 
 $\| \cV_f\|^*(\omega) \in Image \left\{  \cS_{\delta_2}^*  \right\}$ where $\cS_{\delta_2}^*$ is given in \eqref{Aeq-compositions3}. 

On the other hand, $\| \cV_f\|^*$ factors through the map 
$$ \| \cV_f\|^* \colon   H^{n-1}(\| \cL(\lambda_0/10)\| ; \mZ) \to H^{n-1}(\| \cM_{\delta_2}(\delta_3) \| , \mZ) \ .$$
The group $H^{n-1}(\| \cL(\lambda_0/10)\| ; \mZ) \cong \{0\}$, so that $\| \cV_f\|^*(\omega) = 0$, which is a contradiction. 
 \endproof

 
  
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%


 
\end{document}
